

Car dependency in the urban margins: The influence of perceived accessibility on mode choice

Lola Blandin, Giovanni Vecchio, Ricardo Hurtubia, Ignacio Tiznado-Aitken

► To cite this version:

Lola Blandin, Giovanni Vecchio, Ricardo Hurtubia, Ignacio Tiznado-Aitken. Car dependency in the urban margins: The influence of perceived accessibility on mode choice. Transportation Research Part A: Policy and Practice, 2024, 184, pp.104098. 10.1016/j.tra.2024.104098 . hal-04659649

HAL Id: hal-04659649 https://hal.science/hal-04659649

Submitted on 24 Jul 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

1 2	Car dependency in the urban margins: the influence of perceived accessibility on mode choice
3	
4	Lola Blandin ^{a,b} , Giovanni Vecchio ^{c,d} , Ricardo Hurtubia ^{d,e,f} , Ignacio Tiznado-Aitken ^g
5	
6	^a Université Grenoble Alpes, CNRS – INRAE – Grenoble INP, France
7	^b UNCTAD - United Nations Conference on Trade and Development, Geneva, Switzerland
8 9	^c Pontificia Universidad Católica de Chile, Instituto de Estudios Urbanos y Territoriales, El Comendador 1916, Providencia, Santiago, Chile
10 11	^d CEDEUS – Center for Sustainable Urban Development, Pontificia Universidad Católica de Chile, Los Navegantes 1963, Providencia, Santiago, Chile
12 13	^e Pontificia Universidad Católica de Chile, School of Architecture, El Comendador 1916, Providencia, Santiago, Chile
14 15	^f Pontificia Universidad Católica de Chile, Department of Transport Engineering and Logistics, Vicuna Mackenna 4860, Macul, Santiago, Chile
16	⁸ Department of human geography, University of Toronto Scarborough, Toronto, Canada
17	
18 19	Corresponding author: Dr. Blandin Lola, UNCTAD, Palais des Nations, 1201 Geneva, Switzerland <u>lola.blandin@un.org</u>

21 Abstract

22 Car dependence is a dimension of transport poverty whose subjective components have been 23 limitedly explored. Research on car dependence highlights the incidence of transport costs, 24 assesses the multidimensional vulnerability of car-dependency and the possibility to access 25 valued opportunities. However, people's perceptions and their perceived ability to access 26 destinations may better reflect the way they move in car dependent settings. In this paper, we aim to examine what are the determinants of perceived accessibility and to which extent 27 28 perceived accessibility influences mode choices in such areas. Based on a survey carried out in 29 four peripheral and periurban municipalities in the Metropolitan Region of Santiago de Chile, 30 we examine how subjective perceptions of accessibility contribute to explain modal choice in 31 the outskirts. Results show that perceived accessibility has a negative net impact on the utilities 32 for both car and public transport, which means that a low perceived accessibility increases the 33 likelihood of choosing motorized modes. Moreover, residents from peripheral municipalities 34 tend to perceive a higher accessibility than households from periurban areas, who are excluded 35 from the public transport system. These findings show the importance of providing nearby 36 opportunities and convenient alternatives to limit car dependency, especially in periurban areas.

- 39 car dependence
- 40

³⁸ Keywords: discrete choice, mode choice, perceived accessibility, travel behavior, periphery,

1 **1. Introduction**

2 Car dependence can be defined as a situation in which there is a lack of both alternative transport 3 modes to the car and opportunities near residential places, implying that cars become essential 4 to access services, jobs and opportunities (Mattioli, 2016). In car dependent contexts, most 5 forms of transport poverty can be related in one way or another to the dominance of cars over 6 other transport modes, making car dependence a particularly important dimension of transport 7 poverty (Mattioli and Colleoni, 2016). The possibility to own and use a car therefore becomes 8 fundamental in places "built on the assumption of near-universal car access" (Mattioli et al., 9 2020). As a result, in car-dependent areas, those who cannot afford a private vehicle or lack the 10 autonomy to use one may be socially excluded. Others may end up being forced to spend an 11 important share of their income in order to maintain a car (Tiznado-Aitken et al., 2022). The dependence on cars can also contribute to accessibility poverty, that is, the difficulty to reach 12 13 key activities: the car in fact becomes fundamental to access opportunities that are located 14 farther away, implying an expansion of activity spaces and increased travel times and costs 15 (Mattioli, 2021).

16 Transport poverty studies go beyond travel distances, times and expenses dimensions. These 17 include travel expenses, with a focus on fuel (Curl et al., 2018); their relationship with energy 18 and housing costs (Mattioli et al., 2017; Ortar, 2018; Sterzer, 2017); the possibility to engage 19 in various compulsory and non-compulsory activities (Morris, Blumenberg et al., 2020; Morris, 20 Ettema et al., 2020); and the strategies required to access opportunities while coping with 21 transport costs (Belton Chevallier et al., 2018; Tiznado-Aitken et al., 2022). These issues also 22 affect the possibility to reach valued places and opportunities, since moving may imply high 23 monetary and time costs that people may be willing to avoid (Mattioli, 2021). To define this 24 broad concept, Lucas et al.(2016) proposed a multidimensional definition of transport poverty 25 based on five sub-dimensions: mobility poverty, accessibility poverty, transport affordability, 26 exposure to transport externalities and time poverty.

27 Adequate accessibility levels are essential to move towards sustainable mobility patterns 28 (Bertolini et al., 2005), especially in the Latin American context (Valenzuela-Levi et al., 2021; 29 Núñez et al., 2022). Accessibility poverty, which reflects the existing difficulties - in terms of 30 time, ease and costs - in reaching key activities, has been broadly studied and led the 31 development of many indicators and methods to measure it (Vecchio et al., 2020). In particular, 32 accessibility poverty is often explained through time poverty (i.e. when individuals need to 33 spend too much time travelling, leading them to social exclusion or social isolation). In the 34 Global South, time poverty is a particularly relevant concept, especially in peripheral areas, 35 which are often characterized by long travel times and a limited access to cars (Hernandez & Titheridge, 2016). While accessibility is not the only dimension of transport poverty affected 36 37 by inequality, as demonstrates the exposure to transport externalities such as air pollution 38 (Gouveia et al., 2021; 2022) it is especially important considering its contribution to the 39 promotion of social inclusion.

40 While most transport poverty studies focus on objective features (such as travel times or distances), subjective dimensions could also help to explain the forms of accessibility poverty 41 42 experienced in car dependent settings. As the concept of perceived accessibility explains, how individuals rate their environment and their subjective ability to reach a destination and access 43 44 opportunities does not necessarily reflect the results of calculated accessibility evaluations 45 (Lättman et al., 2016; 2018). For example, people's perceptions may question the traditional association between cars and freedom (Sheller & Urry, 2000) while in compact urban settings, 46 the possibility to be car-independent may even generate a feeling of freedom (Lagrell et al., 47 48 2018). In Sweden, a sample of 2711 Malmö residents showed that bicycle users may also

perceive to have better transport accessibility than car drivers or public transport users (Lättman et al., 2018). Yet, subjective dimensions such as perceived accessibility have received less attention, even if various theoretical frameworks on perceived accessibility have pointed out that perceived accessibility is the actual determinant of human spatial behavior (Vecchio & Martens, 2021; Pot et al., 2021).

6 Additionally, research on car dependence has overlooked settings outside the Global North. 7 Until now, most works in fact focused on "developed countries" and their suburban areas 8 (Mattioli et al., 2020) but these issues may be equally important also in other areas of the world. 9 Latin America is a good example in this sense. The region shows high levels of socioeconomic 10 inequality, leading to specific forms of transport poverty that are visible especially in the 11 peripheries of urban areas (Vecchio et al., 2020). Valenzuela-Levi (2018) found in his worldwide study that "more unequal countries, and particularly those in which the rich 12 13 concentrate income, tend to be more car-dependent" (p. 393). Despite having a smaller car 14 ownership rate compared to Global North areas, motorization rates in the region are growing (Yañez-Pagans et al., 2019), determining a higher importance of cars for everyday trips and 15 16 activities. Finally, the rapid expansion of Latin American cities has generated significant forms 17 of suburban development, with significant implications for the metropolitan mobility patterns 18 and the everyday choices of their inhabitants (Blanco & Apaolaza, 2018; Cox & Hurtubia, 2016; 19 Guzman et al., 2017; Obregón Biosca et al., 2018; Tiznado-Aitken et al., 2023). Therefore, in 20 the rapidly growing suburban areas of Latin American cities car dependence may be a further

21 dimension of transport poverty.

Drawing on these premises, we aim to examine to what extent perceived accessibility influences
 modal choices in a car dependent setting. We focus on Santiago de Chile, a Latin American

setting in which the issues of transport poverty have been widely examined, but only partially considering their subjective dimensions. In particular, we intend to compare peripheral and periurban settings, which share a functional dependence from zones of the city in which most opportunities (jobs, health, etc.) are concentrated, but have different socioeconomic conditions and different access to public transport (Tiznado-Aitken et al., 2023). To do so, we examine the results of a survey involving 298 households in four peripheral and periurban municipalities of

30 the Metropolitan Region of Santiago.

31 This study therefore provides several contributions to the existing literature. Given the 32 relatively high socioeconomic inequalities in the Global South, individual perceptions might be 33 better suited to measure inequalities than aggregated objective data. Indeed, current methods 34 based on objective data are not sufficient to identify accessibility poverty in the Global South, 35 in particular in periurban and peripheral areas where transport accessibility indicators are not even available for many municipalities, due to a lack of data in these areas. Therefore, to the 36 37 best of our knowledge, this work is the first to introduce perceived accessibility as an 38 explanatory variable of the mode choice in the Global South context. These results might be 39 particularly relevant to improve transport planning and policies as perceptions might be 40 complementary to objective measures of accessibility. First, the paper discusses the existing mismatching between subjective and calculated accessibility measures by identifying the 41 42 determinants of perceived accessibility in Latin American peripheries. Second, it points out 43 how perceived accessibility might be particularly relevant to explain mode choice process in 44 periurban and peripheral areas.

45 The paper is structured as follows: section 2 gives some highlights on the concept of perceived

46 accessibility, section 3 describes our case study, while section 4 presents the theoretical

47 framework of our modelling approach. Section 5 provides the definition of latent variables and

results, while section 6 concludes by providing key lessons about the role of perceivedaccessibility and future research.

3

4

2. Perceived accessibility and mode choice

5 Accessibility has gained attention as a suitable evaluative approach for assessing the social implications of transport (Lucas, 2012; Martens, 2016; Vecchio et al., 2020), but most studies 6 7 have mainly focused on "objective" and often aggregate measurements, based on variables such 8 as the total number of opportunities and average observed travel times. These objective features 9 may differently affect the opportunities available to each person, considering diverse individual 10 features, aspirations and choices (Vecchio & Martens, 2021). As a result, subjective features should be considered when dealing with accessibility, since people living in the same place may 11 12 take different mobility decisions according to several factors - including perceptions (Van der 13 Vlugt et al., 2019).

14 The concept of perceived accessibility has recently emerged in the transport literature. 15 Perceptions and subjective metrics are relevant since they relate the way people evaluate the conditions in which they live and move (Pacione, 1982). When considering an objective 16 17 territorial measure of accessibility, individuals may perceive accessibility of their residential 18 location differently according to their constraints, needs and feelings. Various theoretical 19 frameworks on perceived accessibility have pointed out that perceived accessibility is the actual 20 determinant of human spatial behavior (and thus mode choice) and that calculated "objective" 21 measures are designed as a proxy for this perception (Pot et al., 2021). Perceived accessibility 22 may therefore influence travel behavior and therefore have impacts on sustainability issues.

23 If perceived accessibility "is about how people rate the conditions in which they live" (Lättman 24 et al., 2018), it might often be confounded with other related concepts, such as perceived 25 convenience, easiness to move, or proximity. Yet, perceived convenience corresponds to a 26 perceived ease of use and perceived usefulness of a service, and therefore includes the idea of 27 utility. Easiness to move is related to what is the easier way to move and proximity describes 28 the nearness in terms of space and time. If these concepts are close, perceived accessibility has 29 the specificity to not only include the idea of proximity but also of being able to reach 30 destinations according to our preferences, which is not necessarily easy.

Comparisons between objective and subjective features showed that the two approaches often identify a low level of agreement and mismatching issues. Different studies have been conducted, focusing on walkability and physical activities (McCormack et al., 2008; Ball et al., 2008; Gebel et al., 2011), access to Metro stations (Shao et al., 2022) existing socio-spatial inequalities related to transport (Curl, 2018; Ryan and Pereira, 2021) or mobility accessibility (Lättman et al., 2018).

37 This mismatching has been explored in relation with different settings. In Australia, Ryan et 38 al., (2016) explored this mismatching in relation with age groups or transport modes. Results 39 showed that older people experienced the lowest accessibility while middle age are associated 40 to the highest. In Sweden, Lättman et al. (2018) showed that levels of calculated "objective" accessibility do not allow predicting perceived accessibility in a given residential area, but also 41 42 highlighted that levels of perceived accessibility differ between main transport modes. 43 Perceived accessibility appears to explain also how people decide to move, but until now, few 44 studies integrated this as an explanatory variable in mode choice modelling. In Hong Kong, 45 Tam et al. (2010) integrated perceived service quality in an airport ground access mode choice model. This variable was defined according to satisfaction level on five selected service 46

1 attributes related to walking time, distance, reliability, travel costs and travel time. Then, in the 2 Netherlands, Scheepers et al. (2016) introduced perceived accessibility to shops, public natural 3 spaces and sports facilities in their neighborhood by car, bicycle and walk in a mode choice 4 model, exploring more specifically the purpose of the trips. Results showed that when perceived 5 accessibility by car was high, persons were more likely to use the car instead of active transport 6 modes. More recently, in Taiwan, a study included includes perceived accessibility as a 7 predictor for behavioral intention based on MaaS users' service experience (Chen & Chen, 8 2023). Yet, these studies were focusing on Global North settings and the latter did not 9 distinguish different residential places.

10

11 **3.** Case study and data

12 <u>3.1. Santiago de Chile: an unequal metropolis</u>

13 The Metropolitan Region of Santiago (SMR), the capital and main city of Chile, hosts more 14 than 7 million inhabitants. The Greater Santiago is composed of 34 autonomous municipalities, 15 but at least 14 outer municipalities are also strongly connected to the urban center and constitute the functional urban area (Fuentes et al., 2018). Santiago and Chile are representative of urban 16 17 trends that can be found in other Latin American countries and more in general in the Global 18 South, including intense forms of urban growth, expansion of peripheral and periurban areas, and the emergence of segregated urban structures (Cox and Hurtubia, 2021; 2022). In Chile, 19 20 almost 90% of the population lives in a city, and 60% of the population lives in the SMR. The 21 periurban areas of Santiago experienced intense demographic growth and land use changes 22 (Banzhaf et al., 2013) in the last few decades, especially outside the urban growth boundaries 23 established in regional plans (Schuster Olbrich et al., 2022), and highways often acted as a push 24 factor for this suburban growth (Borsdorf et al., 2007; Garreton, 2017).

Nonetheless, the case of Santiago has experienced forms of segregation that are stronger - and 25 26 more intentional - than most other areas. Between 1979 and 1985, an eradication policy 27 implemented by the dictatorship displaced families away from the eastside areas that are today occupied by a concentration of high-income residents and the main economic activities (Suazo-28 29 Vecino et al., 2020). As a consequence, the poorest households are often living in low 30 accessibility areas with few job or school opportunities (Tiznado-Aitken et al., 2021), making 31 most work-related trips being directed towards the eastern part of the city, with travel expenses representing a high share of the income for many lower income households. Indeed, the highest 32 33 transport costs (in terms of both time and monetary expenses) are associated to the highly 34 segregated areas of the territory (location, land prices, accessibility), and concentrated in the 35 areas inhabited by the most deprived groups, which tends to reproduce the existing dynamics of socio-spatial inequalities (Iglesias et al., 2019; Moreno Alba et al., 2021; Tiznado-Aitken et 36

37 al., 2022).





Figure 1 Location of the four examined municipalities in the SMR (Source: Authors' own elaboration)

3 The SMR is also affected by governance issues that condition the effectiveness of its transport 4 system. Santiago lacks a metropolitan authority in charge of urban and/or mobility issues, and 5 therefore, the integrated public transport network is available only for the 32 urban 6 municipalities of the city. In our study, we chose to compare four municipalities (see Table 1 7 and Figure 1): two peripherals but included in the public transport system perimeter (Ouilicura and Maipú) and two contiguous periurban municipalities (Colina and Padre Hurtado) that are 8 9 located out of the official city borders. In the latter, independent bus lines exist and allow to 10 connect with the integrated system and thus important hubs located in the center of Santiago. Yet, fares are not integrated, the covered distances are huge and so are the travel times; for 11 12 instance, 32 km separate Colina center from the Costanera Center (approximately the center of 13 mass of the CBD), which represents on average a trip of more than one hour and a half by public 14 transport. Moreover, in peripheral areas, bus stops can be far from residential locations and trips can also be dangerous, for instance if walking at night or close to a highway is necessary to get 15 home (Jirón & Mansilla, 2013). As a result, despite having integrated and frequent public 16 17 transport services, some peripheral areas are characterized by equal or even higher travel times 18 than periurban areas, which are spatially further from the economic center though (Tiznado-19 Aitken et al., 2023).

Table 1 Description of the four municipalities of interest

	Quilicura	Maipú	Padre	Colina
			Hurtado	
Туре	Peripheral	Peripheral	Periurban	Periurban
Population (2017)	210 410	521 627	63 250	146 207
ICVU (2021) ¹	56.23	53.15	49.38	51.49
ICVU CoMo (2021)	67.56	72.05	58.59	50.79
Area	57.5 km ²	135.5 km ²	80.8 km²	971.2 km²
Distance to CBD ²	16.5 km	25.0 km	35.8 km	31.0 km

1

3 Because of all these cumulated difficulties, car-dependency issues are very important in 4 Santiago. These mobility constraints concern everyone, as the inhabitants have to manage 5 complex and expensive travel chains if they want to use combinations of active and public 6 transport. In Santiago, car use considerably increased in the last few years, especially during 7 the pandemic, therefore increasing travel times associated with the congestion costs (Muñoz et 8 al., 2022). However, not everyone can afford its purchase and use, and low-income population 9 is often captive of public transport, while many mid-income households live in car-oriented 10 peripheral zones, under a very high financial and time stress due to depending on their cars for 11 their daily mobility (Cáceres Seguel, 2015). At the same time, high-income households prefer 12 to use a car, as observed in Buenos Aires (Blanco and Apaolaza, 2018) and Santiago (SECTRA, 13 2015). The preference for private vehicles is often justified by the comfort and the cheapness 14 of the vehicles, as cars in Chile have a lower relative and real final price than in the rest of the 15 countries of the region (Figueroa & Rodríguez, 2013). For those who can afford it, results show a high inelasticity compared to the direct cost of its use: despite fuel price increases, there is no 16 observed traffic reduction in the short or medium term (Cortés Salinas et al., 2016). 17

18 <u>3.2. Data collection</u>

19 We used data from a survey developed within the project [*omitted for anonymity reasons*]. The 20 interviews were conducted by phone in the last months of 2021, when some pandemic-related 21 restrictions were still in place, but most in-person activities had resumed. In Santiago, 298 22 residents from the four examined municipalities were asked to describe their trips related to 23 work, educational, shopping and health purposes. Although this is not a representative survey, 24 the panel of respondents shows an acceptable distribution of several key variables of the 25 metropolitan population, particularly in terms of gender and income³. Our dataset highlights 26 the existing differences between periurban and peripheral residents, in terms of age or income 27 (see Table 2).

¹ The ICVU (Indice de Calidad de Vida Urbana, index for urban life quality) measures and compares several life quality index in 99 Chilean cities, covering more than 79% of the population. In particular, the ICVU has a component dedicated to the measurement of "connectivity and mobility" (ICVU CoMo) according to seven variables. The higher is the component; the best are connectivity and mobility and therefore the most accessible is the municipality. See <u>http://icvu.observatoriodeciudades.com/</u> for more information.

² The distance to CBD has been calculated as the distance (by car) between the considered municipality city center and the Costanera Center, often considered as the CBD of Santiago.

³ In Metropolitan Santiago, 51.3% are women, the average household size 3.1, 62% declare to have a job, and there are 15.5 people 65 or older every 100 people between 15 and 64 (INE, 2017). Regarding income

distribution, 30% are up to 2.07 minimum wages, 40% between 2.07 and 4.23 minimum wages and 30% over 4.23 minimum wages (minimum wage reference: 270,000 CLP in 2017) (CASEN, 2020)

Table 2 Descriptiv	e statistics	of the sample
--------------------	--------------	---------------

Observations: 29	8 households		Share (%)	
		Total	Periurban	Peripheral
		(298)	(101)	(197)
Gender	Male	41.3	43.6	40.1
	Female	58.7	56.4	59.9
Age	18 to 30	16.1	15.8	16.2
	30 to 40	20.8	24.8	18.8
	40 to 60	46.6	47.5	46.2
	60 or older	16.5	11.9	18.8
Household size	2 or less	16.9	14.1	18.5
	3	24.1	27.3	22.1
	4	31.0	25.3	35.3
	5 or more	28.0	33.3	24.1
Income (€)	Up to 2 minimum wages	34.5	21.7	40.6
	2 to 4 minimum wages	29.6	25.0	31.8
	More than 4 minimum wages	35.9	53.3	27.6
Number of cars	None	45.4	10.9	23.4
	One	43.0	35.6	51.8
	Two	10.2	34.7	21.8
	Three or more	1.4	18.8	3.0
Employment status	Regular worker	54.6	65.3	57.9
	Student	3.8	3.0	4.6
	Unemployed	19.4	9.9	15.7
	Housekeeper	12.4	10.9	13.7
	Pensioner	9.8	10.9	8.1

3 Moreover, respondents provided information on their perceptions regarding travel behavior and 4 their environment for a set of 15 psychometrics indicators based on 5-levels Likert scale 5 affirmations (see Table 3). These affirmations are related to different transport poverty 6 dimensions (Lucas et al., 2016) such as mobility issues, affordability or monetary expenses, 7 accessibility barriers and easiness to move, and exposure to externalities, including pleasantness 8 of environment, quality of sidewalks or feeling of safety. Answers associated with these 9 affirmations were coded using a 1 to 5 scale. While this question was not designed to explore 10 perceived accessibility in the first place, our hypothesis is that some of these statements could be used to identify indicators of the latent variable "perceived accessibility". 11

Ι	Description	(Strongly)	Neutral	(Strongly)
		disagree		agree
I1	Getting around the city is cheap	65.4	15.1	19.5
I2	You choose the means of transport you use most often because it is the cheapest	46.1	13.2	40.7
I3	You use your usual means of transport because it is the only one available	26.4	15.3	58.3
I4	The locations where you need to go are close to your home	19.9	19.9	60.3
I5	It is easy for you to get to the places you need to go	19.5	16.1	64.4
I6	If you were living somewhere else, it would be easier for you to get to where you need to go	22.9	20.1	57.0
I7	The environment in which you move is pleasant	14.8	15.1	70.1
I8	The environment in which you move has good quality streets and sidewalks	31.0	14.8	54.2
I9	The environment in which you move has good quality cycling infrastructure	54.4	15.8	29.9
I10	Some of your journeys involve other family members	62.0	7.1	31.0
I11	When you leave home, you usually take the opportunity to do several activities before returning home	13.4	11.1	75.5
I12	Other people depend on you to get around the city	46.0	9.1	45.0
I13	You feel insecure because of possible accidents you may suffer when moving around	25.5	15.8	58.7
I14	You have been injured while driving	65.9	8.4	25.7
I15	In general, you feel satisfied when you move	20.5	23.6	55.9

4 Then, respondents provided the destinations of their trips and the transport mode they used (see 5 Table 4), allowing to calculate the associated travel times and distances. These variables were computed using the OpenTripPlanner API⁴ for active modes and car options. For public 6 7 transport, the API was used only to calculate the trips that can be made with the urban public 8 transport system (n=257 trips), since no GTFS files are available for periurban areas in SMR. 9 For travel originating in or directed to these zones, we estimated a travel time connection to Cal 10 v Canto metro station (45 min) for Colina and to Estación Central metro (59 min) for Padre 11 Hurtado. The choice for these connection places has been conditioned by the destination of the 12 few existing bus lines. These estimations come from the travel times declared by the bus companies. We excluded three outlier observations and all the incomplete answers. Each 13 14 respondent may be associated with one or two observations depending on their answers for 15 work and health issues (shopping and educational trips were excluded, as the exact destination address was not provided). In the end, the useful database is composed of 399 complete trip 16 17 observations relating mobility habits, psychometric indicators (perceptions) and socioeconomic 18 attributes. This case study is further analyzed in Blandin (2023) and Hurtubia et al. 19 (forthcoming).

20

Table 4 Modal shares by	residential location
-------------------------	----------------------

	Quilicura (N=119)	Maipú (N=142)	Padre Hurtado (N=62)	Colina (N=76)
Car	48.8%	45.8%	72.6%	78.9%
Public transport	37.0%	41.5%	24.2%	15.8%
Active modes	14.2%	12.7%	3.2%	5.3%

⁴ See <u>https://www.opentripplanner.org/</u> for more information.

4. Methods: defining the integrated choice and latent variable model

1

2 To explain mode choice, the majority of the existing models relies on the use of modal attributes 3 and socioeconomic characteristics (Ben-Akiva & Lerman, 1985), but unobserved factors may 4 have relevant effects on travel behavior, such as perceptions of the environment, accessibility, 5 feasibility. In this sense, the framework of modeling choice behavior was extended with the 6 emergence of the Integrated Choice and Latent Variable (ICLV) model (Ben-Akiva et al., 1999; 7 Walker and Ben-Akiva, 2002). In ICLV models, psychometric indicators are used to measure 8 attitudes and/or perceptions, taking the form of affirmations to which respondents are asked 9 about their level of agreement. These indicators are useful to relate the individual's 10 characteristics with unobserved or latent variables, allowing to build structural relations that 11 can be later used to include the latent variable in a choice model.

- 12 In the literature, latent variables included in ICLV models focusing on mode choice are related 13 to a wide range of concepts, reflecting the multiple dimensions of the decision process. For 14 instance, these variables may be related to lifestyle and or/trip constraints, as individuals may 15 have to face different types of constraints when travelling, whether in terms of duration, location 16 and timing or the need to travel with children (Redmond, 2000). Other variables may be linked 17 with travel liking: individuals may be characterized by positive feelings and perceptions when travelling, which decreases their value of time. For example, some people enjoy their travel 18 19 times in order to work or do other activities (Ory & Mokhtarian, 2005). Safety perception when 20 travelling may explain mode choices (Raveau et al., 2010). For example, the fear of accidents 21 when moving and/or and driving-related injuries may show driving avoidance when not necessary and therefore create a preference for other transport modes. Other studies have also 22 23 included other variables such as comfort, convenience and flexibility (Vredin Johansson et al., 24 2006), social comparison (Abou-Zeid & Ben-Akiva, 2011), environmental concerns (Atasoy et
- al., 2013), values and attitudes (Paulssen et al., 2014) or habitual behaviors (Idris et al., 2015).
- As defined earlier, perceived accessibility reflects individuals' perceptions and feelings related to the ability to reach a destination considering the whole range of personal opportunities and constraints associated to their specific mobility behavior. Therefore, it may be useful to explain individual preferences for choosing a transportation mode.

30 The general structure of the ICLV model is described in Figure 2. Rectangles represent 31 observable variables while ovals represent unobservable and latent variables. Continuous 32 arrows show the structural relations between observable and unobservable variables with the 33 choice model and dotted ones represent the relations between observable variables and the latent variable. In this paper, we use the ICLV model to explain mode choice considering a set 34 35 of three alternatives considering car, public transport or active modes. In our case, the dependent variable is the transport mode chosen by a respondent. Other variables are related to 36 37 the socio-economic characteristics of the households, trip attributes and location characteristics. 38 The psychometric indicators allow to define the latent variable of perceived accessibility (which 39 is unobservable, but defined from indicators and individual characteristics).



 $\frac{1}{2}$

13

3 An ICLV model consists in the combination of two components: a structural equation

4 modelling (SEM) and a discrete choice modelling (DCM). For an individual *n*:

5 The SEM measurement equations that allow describing the relation between the

6 psychometrics indicator and the latent unobservable variable are:

$$I_n = f(X_n^*;\zeta) + v_n \tag{1}$$

8 where the indicator *I* is a function of the latent variable X^* , a set of parameters ζ and an error 9 term (v).

10 The latent variable can also be related to observable explanatory variables X (the

11 characteristics of the household, the residential location). This relation is described by the

12 SEM structural equation:

 $X_n^* = g(X_n; \gamma) + w_n \tag{2}$

14 where γ is a set of parameters and *w* is an error term.

For each transport mode (described as an alternative i, for an individual n), we write a utility function:

17
$$U_{in} = V(X_n, X_i, X_n^*; \beta) + \varepsilon_{in}$$
(3)

18 where *V* is the indirect utility function, β a set of parameters and ε an error term.

In a utility maximization framework, the probability for individual n to choose alternative i in a set of alternative denoted A is:

21
$$P(i|X_n, X_i, X_n^*; \zeta, \sigma_{\varepsilon}) = \operatorname{Prob}[U_{in} \ge U_{jn}, \forall j \in A]$$
(4)

22 where σ_{ε} is a vector of parameters of the error term related to ε .

23 As the latent variable X^* is not observable, it requires to integrate over the distribution of the

24 latent variable. Using the associated density functions, we obtain:

1
$$P(i, I_n | X_n, X_i; \beta, \zeta, \lambda, \sigma_{\varepsilon}, \sigma_v, \sigma_w) = \int_{X_*}^{\Box} P(i | X_n, X_i, X^*; \beta, \sigma_{\varepsilon}) f(I_n | X^*; \zeta, \sigma_v) f(X^* | X_n; \gamma, \sigma_w) dX^* (5)$$

Estimating the parameters implies maximizing the likelihood function of observable variables
 and indicators. The likelihood function (L) is:

$$L = \sum_{n} \sum_{i \text{ in } A} \theta_{in} P(i, I_n | X_n, X_i; \beta, \zeta, \lambda, \sigma_{\varepsilon}, \sigma_{v}, \sigma_{w})$$
(6)

In the literature, many researchers used this approach and applied it to the transport field,
focusing in different issues such as mode choice (Ben-Akiva and Boccara, 1995; Espino et al.,
2006; Hurtubia et al., 2010), car choice (Bolduc et al., 2008), residential location choice
(Walker & Li, 2007) or cycling infrastructures (Rossetti et al., 2018).

9

10 **5. Results**

In this section, we first present the steps followed to identify the latent variable, and thendescribe our results when applying the model.

13 <u>5.1. Identification of the latent variable</u>

We conducted a factor analysis which generated 5 relevant factors with an eigenvalue greater than 1, that together explain 54.19% of the total variance (see Appendix A for more details). We used a Varimax rotation to ease the identification of each variable with a single factor. We did not conduct a previous cluster analysis of the data, as the size of the dataset was not large enough for this purpose. The set of results detailed in Table 5 are limited to indicators associated with an absolute value greater than 0.2 and a 95% significant level.

Table 5 Loading factors associated to the psychometric indicators

I	Description	Factor	Factor	Factor	Factor	Factor
	-	1	2	3	4	5
I1	Getting around the city is cheap					0.366
I2	You choose the means of transport you use most often because it is the cheapest			0.277		0.519
I3	You use your usual means of transport because it is the only one available			0.363	0.247	
I4	The locations where you need to go are close to your home	0.775				
I5	It is easy for you to get to the places you need to go	0.558			0.367	
I6	If you were living somewhere else, it would be easier for you to get to where you need to go	-0.285		0.246		
I7	The environment in which you move is pleasant		0.351		0.275	
I8	The environment in which you move has good quality streets and sidewalks		0.564			
I9	The environment in which you move has good quality cycling infrastructure		0.764			
I10	Some of your journeys involve other family members			0.346		
I11	When you leave home, you usually take the opportunity to do several activities before returning home				0.453	
I12	Other people depend on you to get around the city			0.228		-0.375
I13	You feel insecure because of possible accidents you may suffer when moving around		-0.202	0.440		
I14	You have been injured while driving			0.534	-0.201	0.212
I15	In general, you feel satisfied when you move	0.291			0.429	

By analyzing the corresponding indicators and their signs, it is possible to infer the latent 1 2 variable associated to each factor. Focusing on factor 1, we observe that a high level of 3 agreement with affirmation 4, 5 or 15 demonstrates a feeling of having easy access to 4 opportunities, few difficulties reaching the needed destinations and a positive perception of 5 moving. On the contrary, factor 1 is associated to a disagreement with affirmation 6, meaning 6 that the current residential place appears to be satisfying considering mobility purposes. This 7 shows that factor 1 can be identified as the individual "perceived accessibility" latent variable, 8 defined as "how easy it is to live a satisfactory life with the help of the transport system" 9 (Lättman et al., 2016).

10 Before going any further, we underline that perceived accessibility is a complex issue, which is 11 often mixed up with other similar concepts such as perceived convenience, easiness to move or proximity. Consequently, areas that might objectively be described as having low accessibility 12 13 can still be attractive for residents who perceive it more positively. Despite not being very 14 accessible in terms of travel times and distances, people seem to value the environment in which 15 they live; other elements may therefore explain their residential choices. For instance, some 16 areas may be objectively described as having low accessibility considering all available 17 opportunities, but in reality, households might need a limited amount of close or reachable

18 services to have a high-perceived accessibility.

19 In our specific case, perceived accessibility appears to be particularly related to the close

20 environment. In this sense, exploring in detail the different factors leading to the construction

21 of the perceived latent variable might offer great insights on the differences observed with the

22 Global North context and highlight what is missing in the current accessibility evaluations.

23 Moreover, the inclusion of a satisfaction while traveling item in the definition of our perceived accessibility variable differs from the definition provided by Lättman et al. (2016), which is 24 25 related to satisfaction using a specific mode. It is reasonable to assume that individuals with 26 higher perceived accessibility are likely to be more satisfied with their travel, as this may help 27 bridge the gap between expectations and what is experienced (Stradling, Anable, & Carreno, 28 2007). This relation has been explored in the literature, proving to be complex (Lattman et al., 29 2019; De Vos et al., 2023). For example, lower actual commuting times are associated with 30 higher travel satisfaction, while potential accessibility is not (Pritchard et al., 2021), suggesting 31 again that subjective accessibility indicators are better predictors of behavior than objective ones and that including indicators for travel satisfaction in the measurement of perceived 32 33 accessibility is convenient. Recent efforts have been proposed in this direction, finding a 34 significant discrepancy between standard gravity and satisfaction-based access measures 35 (Chaloux et al., 2019).

A similar analysis performed with factors 2 to 5 allowed to identify other latent variables such as amenities (factor 2), safety (factor 3), and utilitarian travel (factor 5) while factor 4 is not easily identifiable. Yet, we decided to restrict our analysis to the effect of one latent variable in order to focus on the analysis of the potential for perceived accessibility to explain mode choice.

- 40
- 41 <u>5.2. Model specification</u>

42 We specified a logit model including latent variables as defined in section 2. Here, the choice

43 is the transport mode that is used for a trip, considering a set of three alternatives: car, public

44 transport or active modes (bicycle, walk). For each alternative, we defined a utility function:

 $V_{car} = ASC_{car} + \beta_{tt_{car}}TT_{car} + \beta_{cost_{car}}cost_{car} + \lambda_{car} access + \beta_{income_{car}} * income + \beta_{income_{car}} + \beta_{incom$

$$\beta_{peripheral_{car}} * MQ + \sum_{k} \lim \beta_{k_{car}} \delta_k$$
(7.1)

:---:

2
$$V_{pt} = ASC_{pt} + \beta_{tt_{pt}}TT_{pt} + \beta_{cost_{pt}}cost_{pt} + \lambda_{pt} access + \sum_{k} \sum_{k=1}^{k} \beta_{k_{pt}}\delta_{k}$$
(7.2)

3

$$V_{active} = ASC_{active} + \beta_{tt_{active}} TT_{active}$$
(7.3)

4 Where TT variables represent travel times in minutes, *income* is the monthly income described 5 as the number of minimal wages earned monthly, and MQ indicates if the household lives in peripheral municipalities (Maipú or Quilicura). δ_k is a vector of socio-economic characteristics, 6 7 which includes dummy variables for number of cars, gender, presence of children, trip purpose 8 (work or health), and educational level. The latent variable, denoted access, represents the 9 perceived accessibility.

10 The SEM measurement equations are estimated using ordered logit with four thresholds as our 11

psychometrics indicators follow a 5-levels Likert scale. The thresholds are denoted τ_{ii} with i

- 12 within a range from 1 to 4 and i = 4, 5, 6 and 15, corresponding to the 4 most relevant
- 13 psychometrics indicators identified in section 5.1 (Equation (8)).

14

$$I_{j} = \begin{cases} 1 & if \ I_{j}^{*} < \tau_{1j} \\ 2 & if \ \tau_{1j} < I_{j}^{*} < \tau_{2j} \\ 3 & if \ \tau_{2j} < I_{j}^{*} < \tau_{3j} \\ 4 & if \ \tau_{3j} < I_{j}^{*} < \tau_{4j} \\ 5 & if \ I_{i}^{*} > \tau_{4j} \end{cases} \quad \forall \ j = 4, 5, 6, 15$$

$$(8)$$

15 Estimation results for the measurement equation are shown in Appendix B.

16 The latent variable is composed by two variables, corresponding to education and declared

travel time. For an individual n, the structural relation between the latent variable perceived 17

18 accessibility, denoted access, and the socio-economic characteristics is given by:

19
$$access_n = \gamma_{time}(1 + \gamma_{MO} MQ_n)time_n + \gamma_{educ} EDUC_n + w\#(9)$$

20 Where *time* corresponds to the average declared travel time, without distinction on trip purposes 21 or transport mode. It is an approximation of the perceived travel time and is not necessarily 22 realistic, but sufficient to explain perceived accessibility. MQ_n is a dummy variable equals to 1 23 if the respondent lives in Maipú or Quilicura (the peripheral municipalities) and $EDUC_n$ is a 24 dummy variable that indicates if individual *n* has a university degree.

25

5.3. Influence of perceived accessibility in the model 26

27 Results are obtained by using the Apollo package in R (Hess & Palma, 2019). Table 7 provides

28 the estimation results of the ICLV model and a comparison with a logit (MNL) model.

29 The ICLV model allows for the identification of structural relationships between observable 30 and latent variables, which would not be possible using the reduced form choice model. In

31 comparison with the logit model, the latent variable approach allows to remove some

- 1 endogeneity bias and identify additional relevant parameters to explain mode choice (Vij &
- 2 Walker, 2016).

3 All the parameters have the expected signs. We observe that a higher travel time has a negative 4 impact on the utility function of all modes ($\beta_{tt} < 0$). Moreover, earning a high income, having more than a car $(\beta_{nvh_{car}})$ and having a university degree $(\beta_{educ_{car}})$ has a positive effect on the 5 6 utility for car, while being a woman or living in peripheral municipalities has a negative effect. In particular, the utility of car is lower in peripheral municipalities, which is consistent with the 7 8 better overall calculated accessibility observed in this area (as indicated by the ICVU CoMo 9 indicator, see Table 1). This distinction between peripheral and periurban areas was also tested 10 for other modes but was insignificant and therefore removed from the model. Concerning public transport utility, having children has a negative effect while travelling for work 11 purposes $(\beta_{work_{pt}})$ has a positive effect on utility. In the case of work, this can be explained 12 considering the segregated urban structure of Santiago, since most formal job opportunities are 13 concentrated in the affluent eastern area of the city and several subway lines allow to access it. 14 15 Similarly, a network of urban highways connects the eastern zone with the periurban areas here 16 examined. Different is the case for health, since facilities that provide primary attention are 17 usually found locally, while specialist doctors are found mainly in structures located in the 18 central and eastern part of Santiago.

19 We tested the influence of the monetary travel costs on the utility of both motorized modes, 20 which had no significant explanatory power. This could be interpreted as people in this context 21 assuming travel costs as sunken, or unavoidable. In our case, this observation is particularly 22 relevant and can be interpreted as a consequence of the lack of options to travel from periurban 23 and peripheral areas, which strongly affects and conditions the choice of a transport mode. Moreover, the purpose of the trip had not significant explanatory power in the car utility 24 25 function and was excluded from the analysis. When considering the utility function for public transport, the calculated "objective" public transport accessibility of the municipality and the 26 27 residential location were also insignificant.

	Affected utility		ICLV n	nodel	MNL model		
	Vcar	V_{pt}	V_{active}	Value	t-test	Value	t-test
ASC _{car}	Х			0.005	0.08	0.153	0.29
ASC_{pt}		Х		-0.359	-0.46	0.273	0.44
ASC _{bike}			Х	0	-	-	-
$\beta_{tt_{car}}$	Х			-0.053*	-1.70	-0.055*	-1.82
$\beta_{tt_{pt}}$		Х		-0.017*	-1.70	-0.015	-1.57
$\beta_{tt_{bike}}$			х	-0.054***	-3.95	-0.056***	-4.33
$\beta_{cost_{car}}$	Х			-6.97e-5	-0.60	-5.03e-5	1.1e-4
$\beta_{cost_{pt}}$		Х		-1.57e-4	-0.26	-2.86e-5	5.0e-4
$\beta_{incomecar}$	Х			0.214***	3.29	0.203***	3.21
$\beta_{peripheral_{car}}$	Х			-0.711*	-1.74	-0.523	-1.24
$\beta_{nvh_{car}}$	Х			1.386***	4.59	1.362***	4.62
$\beta_{educ_{car}}$	Х			0.720***	2.54	0.714***	2.70
$\beta_{female_{car}}$	Х			-0.590***	-2.26	-0.560***	-2.21
$\beta_{child_{pt}}$		Х		-0.607***	-2.25	-0.506***	-2.00
$\beta_{work_{pt}}$		Х		0.833***	3.09	0.814***	3.14

Table 7 Estimation results of the ICLV and MNL models

λ_{car}	х		-1.220***	-2.79	-	-
λ_{pt}		Х	-1.560***	-3.45	-	-
Ytime	Х	Х	-0.021***	-4.30	-	-
γtime_MQ	Х	Х	-0.321***	-2.04		
Yeduc	Х	Х	-0.540***	-2.79	-	-
LL(final) ⁵			-266.2	3	-278.15	

 $1 \qquad \qquad \\ *** \ p < 0.01, \ ** \ p < 0.05, \ * \ p < 0.10 \\$

The latent variable has a negative impact on the utilities for both car and public transport, which means that a lower perceived accessibility increases the likelihood of choosing these modes, with a slightly more important effect on public transport ($|\lambda_{pt}| > |\lambda_{car}|$). Therefore, a low perceived accessibility increases the probability of choosing to travel by car and even more by public transport. This result is particularly important in terms of transport planning and underlines not only the necessity to provide new mobility options in these territories, but to also explore mechanisms to increase perceived accessibility.

9 Considering the specification of the latent variable, having a university degree has a negative 10 impact on perceived accessibility, which was expected, as a high-level education is generally 11 associated with higher income and a greater value of time. Then, we observe that subjective (or 12 declared) travel times (γ_{time}) have a negative impact on perceived accessibility (i.e. the longer 13 the trip is perceived, the lower is the perceived accessibility), and therefore, declared travel 14 times have an indirect net positive impact on utilities (opposed to the effect of objective travel

15 times).

16 Other socio-economic characteristics such as gender, family size, number of cars, or the

17 objective travel times were not significant to explain the latent variable. In particular, the

18 insignificance of objective travel times in explaining perceived accessibility reinforces the idea

19 that there is no clear correlation between perceived and "objective" accessibility, as existing

20 literature shows (Ryan et al., 2016; Lättman et al., 2018).

21 Finally, households living in peripheral municipalities (Maipú and Quilicura) tend to perceive 22 a higher accessibility, which is consistent with the existing transport supply in these areas. In particular, when comparing the average perceived accessibility in the four municipalities -23 24 computed by considering the average value for the responses in the sample – we observe that 25 Padre Hurtado is perceived as the least accessible (access = -1.31), while Quilicura is seen as 26 the most accessible (-0.60), followed by Maipú (-0.82) and Colina (-0.91). Yet, objective indicators, such as the mobility component of the ICVU indicator (Table 1), classify Maipú as 27 the most and Colina as the less accessible municipalities, indicating again the existing 28 29 mismatching between objective and subjective measures of accessibility.

30

31

6. Discussion and conclusion

In this paper, we investigated the influence of perceived accessibility on the modal choices realized in an unequal, car-dependent setting such as the Santiago Metropolitan Region. In particular, we identified a relevant latent variable to explain the observed mode choice in peripheral and periurban areas.

First, based on the fact that perceptions appear to better explain individuals' behaviors than objective features, it is particularly relevant to integrate subjective indicators when studying

⁵ Final likelihood estimated for each model.

1 mode choice (J. Ryan & Pereira, 2021). In this sense, our results showed that the introduction 2 of perceived accessibility as an explanatory variable in our mode choice model provided a better 3 understanding of the mode choice in peripheral and periurban areas. In particular, our analysis 4 revealed that perceived accessibility in a Latin American context is determined by different 5 subjective features such as the perceived cost of travelling, perceived quality of the environment 6 and/or availability of different transport modes and especially active modes, in line with other 7 studies led in suburban settings (Chowdhury et al., 2016). Many of these features - related to 8 modal options, travel costs and mobility experiences - also appear in studies conducted in the 9 Global North (see for example Friman et al., 2020; Lättman et al., 2019; Pot et al., 2023). It is 10 interesting to observe these commonalities related to perceived accessibility, since these are found in settings that are very different from our case study in terms of size of the urban areas, 11 existing forms of urban inequality and segregation, and limited availability of transport 12 alternatives, all elements that theoretically may differently influence the perception of 13 14 accessibility. Therefore, integrating these new dimensions in future evaluations of accessibility 15 in Latin America may be particularly useful to identify more precisely accessibility poverty.

16 Second, these results are particularly useful for policymakers to tackle car dependency issues by designing equitable public policies and transport planning at the local scale depending on 17 18 individuals' characteristics and perceptions of their environment. While car dependence is of 19 concern from environmental, externalities and climate change standpoints, the consequences 20 are strengthened for those who are forced to use a car, having to bear costs that may lead to debt or a lower quality of life as they spend less on other dimensions of life. The interplay of 21 22 environmental and social concerns requires adopting an intersectional approach, focusing on providing affordable and convenient transportation alternatives as well as opportunities and 23 services in their neighborhoods, especially for those at risk of transport poverty. In this context, 24 25 our results showed that a low perceived accessibility increases the likelihood of choosing to 26 travel by car and even more by public transport in both periurban and peripheral areas. Therefore, increasing residents' (perceived) accessibility is at stake to limit car dependence in 27 28 these areas. Based on these results, providing near opportunities, improving the moving 29 environment and/or developing active mobility infrastructures may have a positive influence 30 on perceived accessibility and may encourage choosing non-motorized modes. This may be 31 especially important in the post-pandemic era, which has shown car dependency intensification 32 as the pandemic spread (Kim et al., 2023) and long-lasting effects remain to be studied.

33 Third, in a Global South context marked by strong socioeconomic inequalities and poor transport planning, the study of perceived accessibility is particularly relevant. In particular, 34 35 perceived accessibility may improve the current measures of accessibility in peripheral and 36 periurban areas, where calculated accessibility is globally low. In Santiago de Chile, periurban 37 areas experienced high levels of urban growth in the last decades, with both high- and middleincome urban inhabitants relocating there looking for features that the city could not offer 38 39 (Cáceres Seguel, 2015; Cox and Hurtubia, 2021). The increasing motorization rates and the 40 relatively recent growth of periurban areas configure car dependency as a newer phenomenon 41 in comparison to Global North countries, as demonstrated also by the scarcity of data and 42 analysis involving settings outside Europe or North America (Saeidizand et al., 2022). Instead, 43 the inhabitants of peripheral areas often did not decide to live in these places and many of them 44 were relocated to the periphery a long time ago, after the eradication of informal settlements located in the eastern affluent areas of the city. Such relocations had important consequences 45 46 also for their everyday mobility, increasing greatly travel times (Suazo-Vecino et al., 2019). 47 Our analysis showed that perceived accessibility reveals that the attractiveness of periurban and 48 peripheral municipalities differs from what could be expected with objective measures. This 49 creates a mismatch between expectations and reality that influences the gap between objective

1 and perceived accessibility. In particular, municipalities that are objectively described as the 2 most accessible, such as Maipú, are perceived as less accessible by residents, who might 3 consider that reaching their destinations considering the current available transport modes set 4 is too complicated, in comparison with Quilicura residents. Moreover, residential self-selection 5 could also be an argument to explain this mismatching, as residents living in the more 6 objective/less perceived accessibility areas might be less interested in the destinations that are 7 included in the objective measures, while those living in more rural areas perhaps are more 8 interested in nearby activities that are nearby in those areas and adapt their travel in line with 9 available services (Pot et al., 2023).

Fourth, satisfaction has been traditionally treated as an outcome of travel experience, which highly varies depending on travel characteristics, choices, and attitudes (De Vos & Witlox, 2017). However, it has been shown that experienced satisfaction can influence travel behavior or even residential location, for example, when users of a transport mode are more likely to locate in areas with higher accessibility for that particular mode (De Vos, 2019). Our results confirm previous results linking travel satisfaction with perceived accessibility (through $\tau_{i.15}$ in Appendix B) providing new evidence on the influence of satisfaction on travel behavior.

17 Fifth, the observations made for mobility costs, which have a neglectable influence on mode 18 choice for every single mode, are interesting and show that mobility choices are based on 19 different concerns in such periurban areas. Undeniably, in these areas people develop residential 20 and mobility choices having other concerns in mind in comparison to other contexts; moreover, 21 specific differences in terms of low carbon behaviors emerge when comparing urban and 22 suburban residents (Neo et al., 2017; Du et al., 2020). In the case of Santiago de Chile, it has 23 been observed that the people who relocate to periurban areas look for bigger houses and 24 neighborhoods safer than those of the central city, and in doing so, they are willing to internalize the increased costs of mobility and access to opportunities (Cáceres Seguel, 2015). This may 25 explain the high car dependency in periurban areas, which may be the only available option for 26 27 traveling large distances given the inadequate level of service offered by public transport. On the contrary, it could be imagined that people who remain in peripheral areas and have access 28 29 to the integrated public transport network may expect to have higher levels of accessibility than 30 those actually available to them, also in comparison to other, more central municipalities of the same city. These attitudes have strong implications in terms of urban planning, underlining the 31 32 necessity to work not only to make existing modal choices more sustainable, but rather adopting 33 a broader definition of accessibility that may include the attitudes, perceptions and feelings of 34 residents regarding mobility environment (Zhang et al., 2021).

35 Yet, this work has some limitations that could lead to future research avenues.

The use of a more extensive dataset, including also other peripheral and periurban municipalities should be explored, as it would allow conducting a cluster analysis before modelling and therefore understanding better the existing differences between peripheral and periurban areas. Moreover, the size of our dataset is limited to 298 respondents. The exclusion of some trip purposes such as shopping is also a limit of this work as it would be of interest to

- 41 study local mobility habits.
- 42 Then, the use of complementary psychometrics indicators may also improve the definition and
- 43 the results associated to the latent variable and other latent constructs that could have emerged
- 44 if better defined. However, we are aware that this would have increased the survey time, which
- 45 makes data collection more difficult and therefore demanded decisions about which purposes
- 46 to prioritize.

- 1 Moreover, this paper is based on an analysis of cross-sectional data. Therefore, this paper gives
- 2 some clues to highlight existing correlations without allowing us to identify causality. Future
- 3 research should particularly focus on understanding the existing causal relations between
- 4 residential choice, mode choice and accessibility.
- 5 Finally, few information is available to assess objective accessibility in periurban municipalities
- 6 and particularly in Colina, which is a limit of the analysis. A complementary work on objective
- 7 accessibility in these specific areas that are poorly documented may be enlightening in order to
- 8 better understand the differences between calculated and perceived accessibility.
- 9
- 10 Acknowledgments: The authors acknowledge the funding received from the Centro de los
- 11 Objetivos de Desarrollo Sostenible para América Latina y el Caribe (CODS). GV and RH also
- 12 acknowledge the research support provided by Centro de Desarrollo Urbano Sustentable
- 13 (CEDEUS), ANID/FONDAP/1523A0004.
- 14 **Funding**: This work was supported by the Centro de los Objetivos de Desarrollo Sostenible
- 15 para América Latina y el Caribe (CODS).
- 16

1 Appendix A

- 2 Table A1 Synthesis of the identified factors, corresponding eigenvalues and explained
- 3 variance.

Factor	Eigenvalue	Explained variance	Cumulative explained variance
1	2.65	17.65	17.65
2	1.79	11.91	29.56
3	1.33	8.90	38.46
4	1.27	8.49	46.95
5	1.08	7.23	54.20

4

5 Appendix B

6 Table B1 – Estimation results for the measurement equation (8).

	Estimate	t-test
$ au_{1.4}$	-5.360	-5.90
$ au_{2.4}^{2,1}$	-3.711	-5.13
$ au_{3,4}^{-,-}$	-2.344	-3.95
$ au_{4,4}$	-0.387	-0.81
$ au_{1,5}$	-5.754	-5.48
$ au_{2,5}$	-4.270	-4.84
$ au_{3,5}$	-2.777	-4.05
$ au_{4,5}$	-0.196	-0.38
$ au_{1,6}$	-1.658	-5.03
$ au_{2,6}$	-0.830	-2.85
$ au_{3,6}$	0.242	0.81
$ au_{4,6}$	1.209	3.60
$ au_{1,15}$	-3.624	-6.22
$ au_{2,15}$	-2.525	-5.18
$ au_{3,15}$	-1.020	-2.79
$ au_{4.15}$	0.532	1.67

1	References
2	Abou-Zeid, M., & Ben-Akiva, M. (2011). The effect of social comparisons on commute well-
3	being. Transportation Research Part A: Policy and Practice, 45(4), 345-361.
4	https://doi.org/10.1016/j.tra.2011.01.011
5	Atasoy, B., Glerum, A., & Bierlaire, M. (2013). Attitudes towards mode choice in
6	Switzerland. disP - The Planning Review, 49(2), 101-117.
7	https://doi.org/10.1080/02513625.2013.827518
8	Ball, K., Jeffery, R. W., Crawford, D. A., Roberts, R. J., Salmon, J., & Timperio, A. F.
9	(2008). Mismatch between perceived and objective measures of physical activity
10	environments. Preventive Medicine, 47(3), 294-298.
11	https://doi.org/10.1016/j.ypmed.2008.05.001
12	Banzhaf, E., Reyes-Paecke, S., Müller, A., & Kindler, A. (2013). Do demographic and land-
13	use changes contrast urban and suburban dynamics? A sophisticated reflection on
14	Santiago de Chile. Habitat International, 39, 179-191.
15	https://doi.org/10.1016/j.habitatint.2012.11.005
16	Belton Chevallier, L., Motte-Baumvol, B., Fol, S., & Jouffe, Y. (2018). Coping with the costs
17	of car dependency : A system of expedients used by low-income households on the
18	outskirts of Dijon and Paris. Transport Policy, 65, 79-88.
19	https://doi.org/10.1016/j.tranpol.2017.06.006
20	Ben-Akiva, M., & Boccara, B. (1995). Discrete choice models with latent choice sets.
21	International Journal of Research in Marketing, 12(1), 9-24.
22	https://doi.org/10.1016/0167-8116(95)00002-J
23	Ben-Akiva, M., & Lerman, S. (1985). Discrete Choice Analysis : Theory and Application to
24	Travel Demand. MIT Press.
25	Ben-Akiva, M., McFadden, D., Gärling, T., Gopinath, D., Walker, J., Bolduc, D., Börsch-
26	Supan, A., Delquié, P., Larichev, O., Morikawa, T., Polydoropoulou, A., & Rao, V.

1	(1999). Extended Framework for Modeling Choice Behavior. Marketing Letters,
2	10(3), 187-203. https://doi.org/10.1023/A:1008046730291
3	Bertolini, L., le Clercq, F., & Kapoen, L. (2005). Sustainable accessibility : A conceptual
4	framework to integrate transport and land use plan-making. Two test-applications in
5	the Netherlands and a reflection on the way forward. Transport Policy, 12(3),
6	207-220. https://doi.org/10.1016/j.tranpol.2005.01.006
7	Blanco, J., & Apaolaza, R. (2018). Socio-territorial inequality and differential mobility. Three
8	key issues in the Buenos Aires Metropolitan Region. Journal of Transport Geography,
9	67, 76-84. https://doi.org/10.1016/j.jtrangeo.2017.07.008
10	Blandin, L. (2023). Residential location and mobility choices: A cross-analysis of
11	vulnerabilities, environmental policies and perceptions. PhD thesis, under the
12	supervision of Mathy S. Université Grenoble Alpes. 227p.
13	Bolduc, D., Boucher, N., & Alvarez-Daziano, R. (2008). Hybrid Choice Modeling of New
14	Technologies for Car Choice in Canada. Transportation Research Record, 2082(1),
15	63-71. https://doi.org/10.3141/2082-08
16	Borsdorf, A., Hidalgo, R., & Sánchez, R. (2007). A new model of urban development in Latin
17	America : The gated communities and fenced cities in the metropolitan areas of
18	Santiago de Chile and Valparaíso. Cities, 24(5), 365-378.
19	https://doi.org/10.1016/j.cities.2007.04.002
20	Cáceres Seguel, C. (2015). Ciudades satélites periurbanas en Santiago de Chile : Paradojas
21	entre la satisfacción residencial y precariedad económica del periurbanita de clase
22	media. Revista INVI, 30(85), 83-110. https://doi.org/10.4067/S0718-
23	<u>83582015000300003</u>

1	Chaloux, N., Boisjoly, G., Grisé, E., El-Geneidy, A., & Levinson, D. (2019). I only get some
2	satisfaction: Introducing satisfaction into measures of accessibility. Transportation
3	research part F: traffic psychology and behaviour, 62, 833-843.
4	Chowdhury, S., Zhai, K., & Khan, A. (2016). The Effects of Access and Accessibility on
5	Public Transport Users' Attitudes. Journal of Public Transportation, 19(1), 97-113.
6	https://doi.org/10.5038/2375-0901.19.1.7
7	Cortés Salinas, A., Figueroa Monsalve, O., & Moreno Alba, D. (2016). Los costos del uso del
8	automóvil y su elasticidad : El caso de Santiago de Chile. Estudios Socioterritoriales,
9	20, 0-0.
10	Cox, T. and Hurtubia, R. (2016). Vectores de expansión urbana y su interacción con los
11	patrones socioeconómicos existentes en la ciudad de Santiago. EURE-Revista de
12	Estudios Urbanos Regionales, 42(127): 185-207.
13	http://www.eure.cl/index.php/eure/article/view/1683
14	Cox, T., & Hurtubia, R. (2021). Subdividing the sprawl : Endogenous segmentation of
15	housing submarkets in expansion areas of Santiago, Chile. Environment and Planning
16	B: Urban Analytics and City Science, 48(7), 1770-1786.
17	https://doi.org/10.1177/2399808320947728
18	Cox, T., & Hurtubia, R. (2022). Compact development and preferences for social mixing in
19	location choices : Results from revealed preferences in Santiago, Chile.
20	https://doi.org/10.1111/jors.12563
21	Curl, A. (2018). The importance of understanding perceptions of accessibility when
22	addressing transport equity : A case study in Greater Nottingham, UK. Journal of
23	Transport and Land Use, 11(1), 1147-1162.

1	Curl, A., Clark, J., & Kearns, A. (2018). Household car adoption and financial distress in
2	deprived urban communities : A case of forced car ownership? Transport Policy, 65,
3	61-71. <u>https://doi.org/10.1016/j.tranpol.2017.01.002</u>
4	De Vos, J., & Witlox, F. (2017). Travel satisfaction revisited. On the pivotal role of travel
5	satisfaction in conceptualizing a travel behaviour process. Transportation research
6	part A: policy and practice, 106, 364-373.
7	De Vos, J. (2019). Satisfaction-induced travel behaviour. Transportation research part F:
8	traffic psychology and behaviour, 63, 12-21.
9	
10	De Vos, J., Lättman, K., Van der Vlugt, A. L., Welsch, J., & Otsuka, N. (2023). Determinants
11	and effects of perceived walkability: a literature review, conceptual model and
12	research agenda. Transport reviews, 43(2), 303-324.
13	Du, M., Cheng, L., Li, X., & Yang, J. (2020). Factors affecting the travel mode choice of the
14	urban elderly in healthcare activity : Comparison between core area and suburban area.
15	Sustainable Cities and Society, 52, 101868. https://doi.org/10.1016/j.scs.2019.101868
16	Espino, R., de Dios Ortúzar, J., & Román, C. (2006). Confidence Interval for Willingness to
17	Pay Measures in Mode Choice Models. Networks and Spatial Economics, 6(2), 81-96.
18	https://doi.org/10.1007/s11067-006-7694-3
19	Figueroa, O., & Rodríguez, C. (2013). Urban Transport, Urban Expansion and Institutions and
20	Governance in Santiago, Chile. Urban Transport.
21	Friman, M., Lättman, K., & Olsson, L. E. (2020). Public transport quality, safety, and
22	perceived accessibility. Sustainability, 12(9), 3563.
23	Friman, M., & Olsson, L. E. (2020). Daily Travel and Wellbeing among the Elderly.
24	International Journal of Environmental Research and Public Health, 17(7), Article 7.
25	https://doi.org/10.3390/ijerph17072342

1	Fuentes, L., Pezoa, M., Fuentes, L., & Pezoa, M. (2018). Nuevas geografías urbanas en
2	Santiago de Chile 1992—2012. Entre la explosión y la implosión de lo metropolitano.
3	Revista de geografía Norte Grande, 70, 131-151. https://doi.org/10.4067/S0718-
4	34022018000200131
5	Garreton, M. (2017). City profile : Actually existing neoliberalism in Greater Santiago. Cities,
6	65, 32-50. https://doi.org/10.1016/j.cities.2017.02.005
7	Gebel, K., Bauman, A. E., Sugiyama, T., & Owen, N. (2011). Mismatch between perceived
8	and objectively assessed neighborhood walkability attributes : Prospective
9	relationships with walking and weight gain. Health & Place, 17(2), 519-524.
10	https://doi.org/10.1016/j.healthplace.2010.12.008
11	Gouveia, N., Kephart, J. L., Dronova, I., McClure, L., Granados, J. T., Betancourt, R. M.,
12	O'Ryan, A. C., Texcalac-Sangrador, J. L., Martinez-Folgar, K., Rodriguez, D., &
13	Diez-Roux, A. V. (2021). Ambient fine particulate matter in Latin American cities :
14	Levels, population exposure, and associated urban factors. Science of The Total
15	Environment, 772, 145035. https://doi.org/10.1016/j.scitotenv.2021.145035
16	Gouveia, N., Slovic, A. D., Kanai, C. M., & Soriano, L. (2022). Air Pollution and
17	Environmental Justice in Latin America : Where Are We and How Can We Move
18	Forward? Current Environmental Health Reports, 9(2), 152-164.
19	https://doi.org/10.1007/s40572-022-00341-z
20	Hernandez, D., & Titheridge, H. (2016). Mobilities of the periphery : Informality, access and
21	social exclusion in the urban fringe in Colombia. Journal of Transport Geography, 55,
22	152-164. https://doi.org/10.1016/j.jtrangeo.2015.12.004
23	Hess, S., & Palma, D. (2019). Apollo : A flexible, powerful and customisable freeware
24	package for choice model estimation and application. Journal of Choice Modelling,
25	32, 100170. https://doi.org/10.1016/j.jocm.2019.100170

1	Hurtubia, R., Atasoy, B., Glerum, A., Curchod, A., & Bierlaire, M. (2010). Considering latent
2	attitudes in mode choice: The case of Switzerland. In World Conference on Transport
3	Research. https://infoscience.epfl.ch/record/152337/
4	Hurtubia, R., Blandin, L., Vecchio, G., Tiznado-Aitken, I. (forthcoming). Accesibilidad
5	percibida y decisiones de movilidad en contextos dependientes del auto. In Rojas, C.,
6	Vecchio, G., Ladrón de Guevara, F. (eds.), Accesibilidad para ciudades equitativas.
7	Santiago de Chile: Ril Editores (pp. 135-164).
8	Idris, A., Habib, K. M. N., Tudela, A., & Shalaby, A. (2015). Investigating the effects of
9	psychological factors on commuting mode choice behavior. Transportation Planning
10	and Technology, 38(3), 265-276. https://doi.org/10.1080/03081060.2014.997451
11	Iglesias, V., Giraldez, F., Tiznado-Aitken, I., & Muñoz, J. C. (2019). How Uneven is the
12	Urban Mobility Playing Field? Inequalities among Socioeconomic Groups in Santiago
13	De Chile. Transportation Research Record, 2673(11), 59-70.
14	https://doi.org/10.1177/0361198119849588
15	Jirón, P., & Mansilla, P. (2013). Atravesando la espesura de la ciudad : Vida cotidiana y
16	barreras de accesibilidad de los habitantes de la periferia urbana de Santiago de Chile.
17	Revista de geografía Norte Grande, 56, 53-74. https://doi.org/10.4067/S0718-
18	34022013000300004
19	Kim, S., Jang, K., & Yeo, J. (2023). Non-linear impacts of COVID-19 pandemic on human
20	mobility : Lessons from its variations across three pandemic waves. Sustainable Cities
21	and Society, 97, 104769. https://doi.org/10.1016/j.scs.2023.104769
22	Lagrell, E., Thulin, E., & Vilhelmson, B. (2018). Accessibility strategies beyond the private
23	car : A study of voluntarily carless families with young children in Gothenburg.
24	Journal of Transport Geography, 72, 218-227.
25	https://doi.org/10.1016/j.jtrangeo.2018.09.002

1	Lättman, K., Friman, M., & Olsson, L. E. (2016). Perceived Accessibility of Public Transport
2	as a Potential Indicator of Social Inclusion. Social Inclusion, 4(3), 36-45.
3	https://doi.org/10.17645/si.v4i3.481
4	Lättman, K., Olsson, L. E., & Friman, M. (2018). A new approach to accessibility –
5	Examining perceived accessibility in contrast to objectively measured accessibility in
6	daily travel. Research in Transportation Economics, 69, 501-511.
7	https://doi.org/10.1016/j.retrec.2018.06.002
8	Lättman, K., Olsson, L. E., Friman, M., & Fujii, S. (2019). Perceived accessibility,
9	satisfaction with daily travel, and life satisfaction among the elderly. International
10	journal of environmental research and public health, 16(22), 4498.
11	Lucas, K. (2012). Transport and social exclusion : Where are we now? Transport Policy, 20,
12	105-113. https://doi.org/10.1016/j.tranpol.2012.01.013
13	Lucas, K., Mattioli, G., Verlinghieri, E., & Guzman, A. (2016). Transport poverty and its
14	adverse social consequences. Proceedings of the Institution of Civil Engineers -
15	Transport, 169(6), 353-365. https://doi.org/10.1680/jtran.15.00073
16	Martens, K. (2016). Transport Justice : Designing fair transportation systems. Routledge.
17	https://doi.org/10.4324/9781315746852
18	Mattioli, G. (2021). Chapter Four - Transport poverty and car dependence : A European
19	perspective. In R. H. M. Pereira & G. Boisjoly (Éds.), Advances in Transport Policy
20	and Planning (Vol. 8, p. 101-133). Academic Press.
21	https://doi.org/10.1016/bs.atpp.2021.06.004
22	Mattioli, G., Lucas, K., & Marsden, G. (2017). Transport poverty and fuel poverty in the UK :
23	From analogy to comparison. Transport Policy, 59, 93-105.
24	https://doi.org/10.1016/j.tranpol.2017.07.007

1	Mattioli, G., Roberts, C., Steinberger, J. K., & Brown, A. (2020). The political economy of
2	car dependence : A systems of provision approach. Energy Research & Social Science,
3	66, 101486. https://doi.org/10.1016/j.erss.2020.101486
4	McCormack, G. R., Cerin, E., Leslie, E., Du Toit, L., & Owen, N. (2008). Objective Versus
5	Perceived Walking Distances to Destinations : Correspondence and Predictive
6	Validity. Environment and Behavior, 40(3), 401-425.
7	https://doi.org/10.1177/0013916507300560
8	Moreno Alba, D. F., Figueroa, Ó., Gurdon, C., Moreno Alba, D. F., Figueroa, Ó., & Gurdon,
9	C. (2021). Desigualdades urbanas : Costos y tiempos de viaje en el Área
10	Metropolitana de Santiago. Revista INVI, 36(102), 54-79.
11	https://doi.org/10.4067/S0718-83582021000200054
12	Morris, E. A., Blumenberg, E., & Guerra, E. (2020). Does lacking a car put the brakes on
13	activity participation? Private vehicle access and access to opportunities among low-
14	income adults. Transportation Research Part A: Policy and Practice, 136, 375-397.
15	https://doi.org/10.1016/j.tra.2020.03.021
16	Morris, E. A., Ettema, D., & Zhou, Y. (2020). Which activities do those with long commutes
17	forego, and should we care? Transportation Research Interdisciplinary Perspectives,
18	5, 100119. https://doi.org/10.1016/j.trip.2020.100119
19	Muñoz, J. C., Vecchio, G., Tiznado-Aitken, I., Guzman, L. A., Arellana, J., & Guimaraes, T.
20	(2022). Transporte y equidad : Abordando la accesibilidad en los margenes urbanos
21	(Policy Report 17). https://cods.uniandes.edu.co/wp-
22	content/uploads/2022/12/DOC_CODS_17.pdf
23	Neo, S. M., Choong, W. W., & Ahamad, R. B. (2017). Differential environmental
24	psychological factors in determining low carbon behaviour among urban and suburban

1	residents through responsible environmental behaviour model. Sustainable Cities and
2	Society, 31, 225-233. https://doi.org/10.1016/j.scs.2017.03.003
3	Núñez, F., Albornoz, E., Gutiérrez, M., & Zumelzu, A. (2022). Socially Sustainable
4	Accessibility to Goods and Services in the Metropolitan Area of Concepción, Chile,
5	Post-COVID-19. Sustainability, 14(21), Article 21.
6	https://doi.org/10.3390/su142114042
7	Ortar, N. (2018). Dealing with energy crises : Working and living arrangements in peri-urban
8	France. Transport Policy, 65, 72-78. https://doi.org/10.1016/j.tranpol.2016.09.008
9	Ory, D. T., & Mokhtarian, P. L. (2005). When is getting there half the fun? Modeling the
10	liking for travel. Transportation Research Part A: Policy and Practice, 39(2), 97-123.
11	https://doi.org/10.1016/j.tra.2004.09.006
12	Otsuka, N., Wittowsky, D., Damerau, M., & Gerten, C. (2021). Walkability assessment for
13	urban areas around railway stations along the Rhine-Alpine Corridor. Journal of
14	Transport Geography, 93, 103081. https://doi.org/10.1016/j.jtrangeo.2021.103081
15	Pacione, M. (1982). The Use of Objective and Subjective Measures of Life Quality in Human
16	Geography. https://doi-org.sid2nomade-1.grenet.fr/10.1177/0309132582006004
17	Paulssen, M., Temme, D., Vij, A., & Walker, J. L. (2014). Values, attitudes and travel
18	behavior : A hierarchical latent variable mixed logit model of travel mode choice.
19	Transportation, 41(4), 873-888. https://doi.org/10.1007/s11116-013-9504-3
20	Pot, F. J., van Wee, B., & Tillema, T. (2021). Perceived accessibility : What it is and why it
21	differs from calculated accessibility measures based on spatial data. Journal of
22	Transport Geography, 94, 103090. <u>https://doi.org/10.1016/j.jtrangeo.2021.103090</u>
23	Pot, F. J., Koster, S., & Tillema, T. (2023). Perceived accessibility and residential self-
24	selection in the Netherlands. Journal of Transport Geography, 108, 103555.

1	Pritchard, J. P., Slovic, A. D., Giannotti, M., Geurs, K., Nardocci, A., Hagen-Zanker, A., &
2	Kumar, P. (2021). Satisfaction with travel, ideal commuting, and accessibility to
3	employment. Journal of Transport and Land Use, 14(1), 995-1017.
4	Raveau, S., Álvarez-Daziano, R., Yáñez, M. F., Bolduc, D., & de Dios Ortúzar, J. (2010).
5	Sequential and Simultaneous Estimation of Hybrid Discrete Choice Models : Some
6	New Findings. Transportation Research Record, 2156(1), 131-139.
7	https://doi.org/10.3141/2156-15
8	Redmond, L. (2000). Identifying and Analyzing Travel-Related Attitudinal, Personality, and
9	Lifestyle Clusters in the San Francisco Bay Area.
10	https://escholarship.org/uc/item/0317h7v4
11	Rossetti, T., Guevara, C. A., Galilea, P., & Hurtubia, R. (2018). Modeling safety as a
12	perceptual latent variable to assess cycling infrastructure. Transportation Research
13	Part A: Policy and Practice, 111, 252-265. https://doi.org/10.1016/j.tra.2018.03.019
14	Ryan, J., & Pereira, R. H. M. (2021). What are we missing when we measure accessibility?
15	Comparing calculated and self-reported accounts among older people. Journal of
16	Transport Geography, 93, 103086. https://doi.org/10.1016/j.jtrangeo.2021.103086
17	Ryan, M., Lin, T., Xia, J. (Cecilia), & Robinson, T. (2016). Comparison of perceived and
18	measured accessibility between different age groups and travel modes at Greenwood
19	Station, Perth, Australia. European Journal of Transport and Infrastructure Research,
20	16(2), 406-423.
21	Saeidizand, P., Fransen, K., & Boussauw, K. (2022). Revisiting car dependency : A
22	worldwide analysis of car travel in global metropolitan areas. Cities, 120, 103467.
23	https://doi.org/10.1016/j.cities.2021.103467
24	Scheepers, C. E., Wendel-Vos, G. C. W., van Kempen, E. E. M. M., de Hollander, E. L.,
25	van Wijnen, H. J., Maas, J., den Hertog, F. R. J., Staatsen, B. A. M., Stipdonk, H. L.,

1	Int Panis, L. L. R., van Wesemael, P. J. V., & Schuit, A. J. (2016). Perceived
2	accessibility is an important factor in transport choice—Results from the AVENUE
3	project. Journal of Transport & Health, 3(1), 96-106.
4	https://doi.org/10.1016/j.jth.2016.01.003
5	Schuster Olbrich, J. P., Vich, G., Miralles-Guasch, C., & Fuentes, L. (2022). Urban sprawl
6	containment by the urban growth boundary : The case of the Regulatory Plan of the
7	Metropolitan Region of Santiago of Chile. Journal of Land Use Science, 17(1),
8	324-338. https://doi.org/10.1080/1747423X.2022.2086312
9	SECTRA. (2015). Encuesta Origen Destino Gran Santiago. Universidad Alberto Hurtado,
10	Observatorio Social.
11	Shao, R., Derudder, B., & Yang, Y. (2022). Metro accessibility and space-time flexibility of
12	shopping travel : A propensity score matching analysis. Sustainable Cities and
13	Society, 87, 104204. https://doi.org/10.1016/j.scs.2022.104204
14	Sheller, M., & Urry, J. (2000). The City and the Car. International Journal of Urban and
15	Regional Research, 24(4), 737-757. https://doi.org/10.1111/1468-2427.00276
16	Sterzer, L. (2017). Does competition in the housing market cause transport poverty?
17	Interrelations of residential location choice and mobility. European Transport
18	Research Review, 9(3), 45. https://doi.org/10.1007/s12544-017-0259-3
19	Stradling, S. G., Anable, J., & Carreno, M. (2007). Performance, importance and user
20	disgruntlement: A six-step method for measuring satisfaction with travel
21	modes. Transportation Research Part A: Policy and Practice, 41(1), 98-106.
22	Suazo-Vecino, G., Muñoz, J. C., & Fuentes Arce, L. (2020). The Displacement of Santiago de
23	Chile's Downtown during 1990–2015 : Travel Time Effects on Eradicated Population.
24	Sustainability, 12(1), Article 1. https://doi.org/10.3390/su12010289

1	Tam, M. L., Lam, W. H. K., & Lo, H. P. (2010). Incorporating passenger perceived service
2	quality in airport ground access mode choice model. <i>Transportmetrica</i> , 6(1), 3-17.
3	https://doi.org/10.1080/18128600902929583
4	Tiznado-Aitken, I., Lucas, K., Muñoz, J. C., & Hurtubia, R. (2022). Freedom of choice?
5	Social and spatial disparities on combined housing and transport affordability.
6	Transport Policy, 122, 39-53. https://doi.org/10.1016/j.tranpol.2022.04.005
7	Tiznado-Aitken, I., Muñoz, J. C., & Hurtubia, R. (2021). Public transport accessibility
8	accounting for level of service and competition for urban opportunities : An equity
9	analysis for education in Santiago de Chile. Journal of Transport Geography, 90,
10	102919. https://doi.org/10.1016/j.jtrangeo.2020.102919
11	Tiznado-Aitken, I., Vecchio, G., Guzman, L. A., Arellana, J., Humberto, M., Vasconcellos,
12	E., & Muñoz, J. C. (2023). Unequal periurban mobility : Travel patterns, modal
13	choices and urban core dependence in Latin America. Habitat International, 133,
14	102752. https://doi.org/10.1016/j.habitatint.2023.102752
15	Valenzuela-Levi, N. (2018). Why do more unequal countries spend more on private vehicles?
16	Evidence and implications for the future of cities. Sustainable Cities and Society, 43,
17	384-394. https://doi.org/10.1016/j.scs.2018.09.003
18	Valenzuela-Levi, N., Echiburu, T., Correa, J., Hurtubia, R., & Muñoz, J. C. (2021). Housing
19	and accessibility after the COVID-19 pandemic : Rebuilding for resilience, equity and
20	sustainable mobility. Transport Policy, 109, 48-60.
21	https://doi.org/10.1016/j.tranpol.2021.05.006
22	Van der Vlugt, AL., Curl, A., & Wittowsky, D. (2019). What about the people? Developing
23	measures of perceived accessibility from case studies in Germany and the UK. Applied
24	Mobilities, 4(2), 142-162. https://doi.org/10.1080/23800127.2019.1573450

1	Vecchio, G., & Martens, K. (2021). Accessibility and the Capabilities Approach : A review of
2	the literature and proposal for conceptual advancements. Transport Reviews, 41(6),
3	833-854. https://doi.org/10.1080/01441647.2021.1931551
4	Vecchio, G., Tiznado-Aitken, I., & Hurtubia, R. (2020). Transport and equity in Latin
5	America : A critical review of socially oriented accessibility assessments. Transport
6	Reviews, 40(3), 354-381. https://doi.org/10.1080/01441647.2020.1711828
7	Vij, A., & Walker, J. L. (2016). How, when and why integrated choice and latent variable
8	models are latently useful. Transportation Research Part B: Methodological, 90,
9	192-217. https://doi.org/10.1016/j.trb.2016.04.021
10	Vredin Johansson, M., Heldt, T., & Johansson, P. (2006). The effects of attitudes and
11	personality traits on mode choice. Transportation Research Part A: Policy and
12	Practice, 40(6), 507-525. https://doi.org/10.1016/j.tra.2005.09.001
13	Walker, J., & Ben-Akiva, M. (2002). Generalized random utility model. Mathematical Social
14	Sciences, 43(3), 303-343. https://doi.org/10.1016/S0165-4896(02)00023-9
15	Walker, J. L., & Li, J. (2007). Latent lifestyle preferences and household location decisions.
16	Journal of Geographical Systems, 9(1), 77-101. https://doi.org/10.1007/s10109-006-
17	0030-0
18	Yañez-Pagans, P., Martinez, D., Mitnik, O. A., Scholl, L., & Vazquez, A. (2019). Urban
19	transport systems in Latin America and the Caribbean : Lessons and challenges. Latin
20	American Economic Review, 28(1), 15. https://doi.org/10.1186/s40503-019-0079-z
21	Zhang, R., Zhang, J., Long, Y., Wu, W., Liu, J., & Jiang, Y. (2021). Long-term implications
22	of electric vehicle penetration in urban decarbonization scenarios : An integrated land
23	use-transport-energy model. Sustainable Cities and Society, 68, 102800.
24	https://doi.org/10.1016/j.scs.2021.102800